\documentclass{article}

\usepackage[top=0.5in,left=1in,right=1in,bottom=0.5in]{geometry}
\usepackage{hyperref}

\pagenumbering{gobble}



\title{EECS 210 - Discrete Structures \\ \Large Discussion Session}
\author{Instructor: Dr. Matthew Moore \\ Grauate Teaching Assistant: Andrew Stratmann}
\date{\today}

\begin{document}
	
\maketitle

\section*{About Me - Office Hours \& Contact Information}

Name: Andrew Stratmann \\
Office: 2041 Eaton \\
Email: adstratm@ku.edu \\
\textbf{Discussion Sections:}
\begin{tabbing}
	\hspace{30pt}\=\hspace{60pt}\=\hspace{80pt}\=\kill
	\> 2000 \> W 4:30 - 6:20 \> Learned 1136\\
	\> 1600 \> TH 4:30 - 6:20 \> Learned 2133
\end{tabbing}



\noindent \textbf{Office Hours:}
\begin{tabbing}
\hspace{30pt}\=\hspace{60pt}\=\kill
\> Wednesday: \> 11:00 AM - 12:00 PM\\
\> Thursday: \> 11:00 AM - 12:00 PM
\end{tabbing}

I am also available by appointment and by email.  To make an appointment, send me an email or ask me in discussion.


\section*{Discussion Session Structure}

Each discussion section will last one hour and fifty minutes and will generally contain the following elements:

\begin{enumerate}
\item \textbf{Quiz} - Each week you will be given a quiz over past materials.
\item \textbf{Quiz Review} - After the quizzes are collected I will go over the solutions.
\item \textbf{Homework Help} - I will provide hints and assistance on homework problems you bring to class.  Be sure to come to discussion with any problems requiring assistance.
\end{enumerate}

\section*{Attendance}

Attendance will not be recorded but students arriving after quiz solutions have been presented will not be permitted to take the quiz.

\end{document}